Author's response to reviews

Title: Disparities in bone density measurement history and osteoporosis medication utilisation in Switzerland: Results from the Swiss Health Survey 2007

Authors:

Rita Born (born.mph@bluewin.ch)
Marcel Zwahlen (zwahlen@ispm.unibe.ch)

Version: 2 Date: 21 March 2012

Author's response to reviews:

We now state at the end of the methods section: “Ethical approval was not required for this analysis of data which is freely available to research institutions.”